Author's response to reviews

Title: Unusual Causes of Parkinsonism: a case series of potential cause-effect relationship

Authors:

Solomon O Ugoya Dr (docsouls@yahoo.com)
Emmanuel I Agaba Dr (eiagaba@yahoo.com)
Comfort A Daniyam Dr (comfort_13@yahoo.com)

Version: 4 Date: 3 October 2010

Author's response to reviews: see over
The Editor,

JMC

Attention: Editorial Team

RE: MS: 6129007063691597- Submission of Manuscript titled: Unusual causes of parkinsonism: a case series

We are submitting the corrected version of our paper for possible publication in your reputable journal.

The suggested corrections are as follows: to change title to; Unusual causes of parkinsonism: a case series of potential cause-effect relationship.

- All the necessary typographical errors are made and changes tracked.
- The discussions has been re-structured and referenced appropriately.
- 3 new referenced added and discussed.
- One reviewer suggested a change of name case series as there are two cases. This we declined because is not within the journal definition of case series.
- The dose of quinine the patient to in case #1 could not be ascertained, though she claimed it was up to five tablets (which mean about 750mg).
- In conclusion, References has been reviewed and changed, from 12 to 17.
- Thank you for your effort in reading our manuscript.

Yours sincerely,

Dr. Solomon O. Ugoya